Author’s response to reviews

Title: C-reactive protein, established risk factors and social inequalities in cardiovascular disease -- the significance of absolute versus relative measures of disease

Authors:

Maria Rosvall (Maria.rosvall@med.lu.se)
Gunnar Engstrom (Gunnar.engstrom@med.lu.se)
Goran Berglund (Goran.berglund@med.lu.se)
Bo Hedblad (Bo.hedblad@med.lu.se)

Version: 8 Date: 6 May 2008

Author’s response to reviews:

We have corrected some references and some words in the text.